Author's response to reviews

Title: Closing-wedge high tibial osteotomy: survival and risk factor analysis at long-term follow up

Authors:

Turgay Efe (efet@med.uni-marburg.de)
Gafar Ahmed (ortopae@med.uni-marburg.de)
Thomas j Heyse (heyse@med.uni-marburg.de)
Ulrich Boudriot (orthopaedie@sankt-elisabeth-hospital.de)
Nina Timmesfeld (timmesfe@med.uni-marburg.de)
Susanne Fuchs-Winkelmann (orthopae@med.uni-marburg.de)
Bernd Isaque (orthopae@med.uni-marburg.de)
Markus D Schofer (schofer@med.uni-marburg.de)

Version: 2 Date: 24 September 2010

Author's response to reviews: see over
Dear Editor,

thank you very much for your reply.

**Ethics**
The study was authorised by the local ethical committee. All patients gave informed consent to participate in the study. This project was performed in accordance to the Helsinki Declaration, and to local legislation.

**Acknowledgements**
We wish to thank Prof. Michael Pfeiffer and Prof. Peter Griss for performing the HTO procedure.

We added a paragraph in the methods and acknowledgements section stating this.

Best wishes,

Turgay Efe